Author's response to reviews

Title: Assessing the context of health care utilization in Ecuador: A spatial and multilevel analysis

Authors:

Daniel F Lopez-Cevallos (lopezced@wou.edu)
Chunhuei Chi (chunhuei.chi@oregonstate.edu)

Version: 4 Date: 9 March 2010

Author's response to reviews:

Assessing the context of health care utilization in Ecuador:
A spatial and multilevel analysis.

Authors’ responses to third revision (formatting)

Our specific responses are included below. Each answer starts with AR: (authors’ response).

Major revisions (we require the author to make these changes)

Footnotes: We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that you incorporate the footnotes into the text of the manuscript. Alternatively, you may wish to make an appendix section containing your footnotes.
AR: We have incorporated the footnotes into the text of the manuscript.

Tables: tables should fit into an A4 page (maximum size 176mm x 230mm) and appear in portrait format.
If you cannot re-orientate the table to fit this size, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.
AR: We have fitted all tables into an A4 page and appear in portrait format.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Placement text: Please remove all figure and table placement text.
AR: We have removed all figure and table placement text.

Typography: Please take this opportunity to check your manuscript for any
typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

AR: We have checked the manuscript for typographical errors and made final corrections.